Title: Case report of non-secretory multiple myeloma expressed as multiple extramedullary plasmacytoma with an endobronchial lesion mimicking metastatic cancer

Reviewer #1:

Scientific Quality: Grade B (Very good)

Language Quality: Grade B (Minor language polishing)

Conclusion: Minor revision

Specific Comments to Authors: The authors reported a patient of non-secretory multiple myeloma expressed as multiple extramedullary plasmacytoma with an endobronchial lesion. They stated that although several cases of plasmacytoma in the trachea have been reported, most cases are solitary or progressive MM, and this case is extremely rare. This finding may be of interest to pulmonologists and hematologists. I have the following concerns. 1. In the Treatment section, it would be helpful to show computed tomography images after radiotherapy to understand the clinical course better, if available. 2. There are no clear staging criteria for non-secretory multiple myeloma. It would be easier to understand the severity of this case if albumin and beta-2 microglobulin were shown as a reference.

We thank you for your insightful and valuable comments.

Answer

1. To better understand the clinical course, computed tomography images after radiation therapy were added to the paper.

2. Albumin and beta-2 microglobulin level were added and International Staging System stage at the time of diagnosis was II.
Science editor:

The author present a rare case of non-secretory multiple myeloma with multiple extramedullary plasmacytoma with an endobronchial lesion. The manuscript is ok, but the case data need to be improved, and some language errors need to be corrected.

Language Quality: Grade C (A great deal of language polishing)
Scientific Quality: Grade B (Very good)

Answer:

The data and contents of the case were added to the main text, and English correction was re-executed. The reviewer's comments were also corrected.

Sincerely,

Seul Bi Lee, MD